\begin{table}%tA.1
%\centering
\par
\caption{Large Magellanic Cloud.}
\begin{tabular}{ccccl}
\hline\hline
\noalign{\smallskip}
Freq.     & Wavel.       & \multicolumn{2}{c}{Flux densities} \\
 $\nu$   & $\lambda$  &  Jy & $\Delta$Jy  & Reference\\
(GHz)    & (mm)          &\multicolumn{3}{c}{}\\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
0.0197  &  15~230       & 5270   &  1054	   & Shain (1959)\tablefootmark{a} \\
0.045   &   6670       & 2997   &  450     & Alvarez et~al. (\cite{Alvarez87}) \\
0.0855  &   3510       & 3689   & 400	   & Mills (1959)\tablefootmark{a}  \\
0.0968  &   3100       & 2839   & 600	   & Mills (1959)\tablefootmark{a} 	\\ 
0.158   &   1900       & 1736   & 490	   & Mills (1959)\tablefootmark{a}  \\
0.408   &    735       & 925    & 30	   & Klein et~al. (\cite{Klein89}) \\
0.408   &    735       & 934    & 173      & Haslam et~al. (1981)\tablefootmark{a}  \\ 
1.4     &    214       & 529    & 30	   & Klein et~al. (\cite{Klein89}) \\
1.4     &    214       & 450    & 30	   & Hughes et~al. (\cite{Hughes06})	\\ 
2.3     &    130       & 404    & 30	   & Klein et~al. (\cite{Klein89}) \\
2.3     &    130       & 412    & 50       & Mountfort et~al. (\cite{Mountfort87}) \\
2.45    &    122       & 390    & 20       & Haynes et~al. (\cite{Haynes91}) \\
4.75    &     63       & 363    & 30       & Haynes et~al. (\cite{Haynes91}) \\
4.75    &     63       & 296    & 25	   & Hughes et~al. (\cite{Hughes06}) \\
8.55    &     35       & 270    & 35       & Haynes et~al. (\cite{Haynes91}) \\
23	&     13       & 160	& 25	   & This paper (2010) \\
33	&     9.1      & 159	& 25	   & This paper (2010) \\
41	&     7.3      & 165	& 25	   & This paper (2010) \\
61      &     4.9      & 205	& 32	   & This paper (2010) \\
93	&     3.2      & 366	& 65	   & This paper (2010) \\
245	&     1.2      & 1630   & 170	   & Aguirre et~al. (\cite{Aguirre03}) \\
400	&    0.75      & 7930   & 590	   & Aguirre et~al. (\cite{Aguirre03}) \\ 
460	&    0.65      & 10~570  &  890	   & Aguirre et~al. (\cite{Aguirre03}) \\
630	&    0.48      & 29~660  & 2980	   & Aguirre et~al. (\cite{Aguirre03}) \\
1250    &    0.24      & 112~300 & 2400	   & Aguirre et~al. (\cite{Aguirre03}) \\
1250    &    0.24      & 153~000 & 3000 	   & This paper (2010) \\
2140    &    0.14      & 177~700 & 3800	   & Aguirre et~al. (\cite{Aguirre03}) \\
2140    &    0.14      & 251~100 & 4500 	   & This paper (2010) \\	
3000    &    0.10      & 130~100 & 17~600    & Aguirre et~al. (\cite{Aguirre03}) \\\
3000    &    0.10      & 200~200 & 3000	   & This paper (2010) \\
3000    &    0.10      & 206~250 & 30~000    & Hughes et~al. (\cite{Hughes06}) \\
3000    &    0.10      & 184~687 & 28~000	   & Rice et~al. (\cite{Rice88}) \\
5000    &    0.06      & 82~917  & 12~000	   & Rice et~al. (\cite{Rice88}) \\
5000    &    0.06      & 50~500 	& 7500     & This paper (2010) \\
5000    &    0.06      & 96~250  & 14~000    & Hughes et~al. (\cite{Hughes06}) \\
12~000   &    0.025     & 7824   & 1200	   & Rice et al. (\cite{Rice88}) \\
12~000   &    0.025     & 7520  	& 1100	   & This paper (2010) \\
25~000   &    0.012     & 2782   & 400	   & Rice (\cite{Rice88}) \\
25~000   &    0.012     & 3450   & 600      & This paper (2010) \\
61~220   &   0.0049     & 1240	& 190      & This paper (2010) \\
85~710	&   0.0035     & 2190	& 300      & This paper (2010) \\
136~365  &   0.0022     & 3765	& 400      & This paper (2010) \\
240~000  &   0.00125    & 4520	& 650      & This paper (2010) \\	
542~000	&   0.00055    & 2520   & 139      & de Vaucouleurs et~al. (\cite{deVaucouleurs91}) \\
681~000	&   0.00044    & 1840	&  87      & 
de Vaucouleurs et~al. (\cite{deVaucouleurs91}) \\
819~000	&   0.000365   & 783	&  53      & de Vaucouleurs et~al. (\cite{deVaucouleurs91}) \\
2~142~850 &   0.00014    & 204	&  40      & Page \& Carruthers (\cite{Page81}) \\
2~307~700 &   0.00013    & 228	&  45      & Page \& Carruthers (\cite{Page81}) \\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
\end{tabular}
\tablefoot {\tablefoottext{a}{As revised and listed by Klein  et~al. (\cite{Klein89}).}}
\end{table}